Author’s response to reviews

Title: Antibiotics and antibiotic-resistant bacteria in waters associated with a hospital in Ujjain, India

Authors:

Vishal Diwan (vishaldiwan@hotmail.com)
Ashok J Tamhankar (ajtamhankar@yahoo.com)
Rakesh K Khandal (rkhandal@shriraminstitute.org)
Shanta Sen (shanta@shriraminstitute.org)
Manjeet Aggarwal (manjeet@shriraminstitute.org)
Yogyata Marothi (dryogyata@yahoo.com)
Rama V Iyer (mohan_bhopal@yahoo.co.in)
Karin Sundblad-Tonderski (karsui@ifm.liu.se)
Cecilia Stålsby-Lundborg (Cecilia.Stalsby.Lundborg@ki.se)

Version: 4 Date: 13 July 2010

Author’s response to reviews:

Covering Letter July 13 2007
To,
The BMC Series Editorial Production Team

I have now checked the MS for all major and minor revisions

1. The list of authors in the manuscript is now written exactly as they are in the submission system, both in style and order.

2. Abstract present in the manuscript matches exactly what is in the submission system

3. Provided functioning link to the website for reference 17

4. Removed all unnecessary underlines

5. Checked manuscript for typographical errors

6. Page 2, Line 6th and page 18 line 4th - ‘KT’ is changed to ‘KST’

7. Page 4 Background second paragraph - ‘It is not yet very clear whether the presence of antibiotics in the aquatic environment also results in the bacteria found there developing resistance’ is reframed as ‘It is not yet very clear whether the presence of antibiotics in the aquatic environment results in developing resistance in the bacteria found there.’

Thanks and regards
Vishal Diwan
Corresponding Author
vishaldiwan@hotmail.com